Author’s response to reviews

Title: Granular cell tumor presenting as an intraocular mass: a case report

Authors:

Jingyuan Yang (yangjingyuan90@126.com)

Xinyu Ren (renxinyupumch@126.com)

Youxin Chen (chenyx@pumch.cn)

Version: 2 Date: 15 Mar 2019

Author's response to reviews:

We would like to express our sincere thanks to the reviewers for the detailed, constructive and positive comments.

Replies to Reviewer 1

No further suggestions.

Replies to Reviewer 2

1. In Figure 1E, no green arrowhead is shown.

Response: We deeply apologize for this mistake, and we thank Dr. Kusuhara very much for this suggestion. We add a green arrowhead in the part (E) of the Figure, which can be found in the revised Figure.

2. The legend of Figure 1F should be amended (e.g., (F) Positive immunostaining for S-100 was confirmed.).

Response: We thank Dr. Kusuhara very much for this comment. We have amended the legend of Figure 1F as Dr. Kusuhara suggested, which can be found in the part of Figure legends at the end of the revised manuscript.